Author's response to reviews

Title: Non-adherence to WHO recommended inter-birth intervals in Rufiji, Tanzania

Authors:

Amon Exavery (amonexavery@yahoo.com)
Sigilbert Mrema (smrema@ihi.or.tz)
Amri Shamte (ashamte@ihi.or.tz)
Kristin Bietsch (kbietsch@princeton.edu)
Dominic Mosha (dmosha@ihi.or.tz)
Godfrey Mbaruku (gmbaruku@ihi.or.tz)
Honorati Masanja (hmasanja@ihi.or.tz)

Version: 2 Date: 24 July 2012

Author's response to reviews: see over
Executive Editor,
Emily Crow, PhD,
BMC Pregnancy & Child Birth,

Re-submission of a revised manuscript “MS 1775102521766358: Non-adherence to WHO recommended inter-birth intervals in Rufiji, Tanzania” for publication in the BMC Pregnancy & Child Birth

The manuscript above has been revised as the Executive Editor advised. Specifically, we added a sentence indicating that permission to access the data was granted by the authors’ institution which also owns the data source. This sentence appears under ‘Data and study population’ section of the revised manuscript that has also been uploaded on the journal’s website.

We hope our revision responds fully to the Editor’s concern, and where this is not the case, we are comfortable and willing to undertake further revisions as the Editor may further advise.

Sincerely,

Amon Exavery,
For the authors